Author's response to reviews

Title: Intraluminal instillation of urokinase and autologous plasma: A method to unblock occluded central venous ports

Authors:

Georg Seifert (georg.seifert@charite.de)
Hanno Riess (hanno.riess@charite.de)
Karl H Seeger (karl.seeger@charite.de)
Guenter Henze (guenter.henze@charite.de)
Anja Borgmann (ania.borgmann@charite.de)

Version: 5 Date: 6 April 2006

Author's response to reviews:

We considered all the suggestions of the editor.

Sincerely
Georg Seifert